Author’s response to reviews

Title: Daylight photodynamic therapy versus cryosurgery for the treatment and prophylaxis of actinic keratoses of the face – Protocol of a multicenter, prospective, randomized, controlled, two-armed study

Authors:

Elisabeth Datz (dr.elisabeth.kohl@gmail.com)
Michael Koller (michael.koller@ukr.de)
Florian Zeman (florian.zeman@ukr.de)
Rolf-Markus Szeimies (rolf-markus.szeimies@ukr.de)
Wolfgang Philipp-Dormston (w.dormston@hotmail.com)
Welf Prager (welf.prager@icloud.de)
Peter Gerber (PeterArne.Gerber@med.uni-duesseldorf.de)
Sigrid Karrer (sigrid.karrer@klinik.uni-regensburg.de)

Version: 1 Date: 26 Jun 2017

Author’s response to reviews:

Dear Dr. Tu,

I forwarded copies of all ethical approvals and an English translation to the following e-mail address: BMC Series Editorial@biomedcentral.com.

This is an ongoing clinical trial. This sentence was also included in the methods section.

Yours sincerely

Sigrid Karrer